Author's response to reviews

Title: Prompt improvement of a pressure ulcer by the administration of high viscosity semi-solid nutrition via a nasogastric tube in a man with tuberculosis: a case report

Authors:

Tamaki Nakayama (shayashi@kch.hosp.go.jp)
Seiji Hayashi (shayashi@kch.hosp.go.jp)
Kyoichi Okishio (shayashi@kch.hosp.go.jp)
Tomoko Tomishiro (shayashi@kch.hosp.go.jp)
Kaori Hosogai (shayashi@kch.hosp.go.jp)
Yuki Ootsu (shayashi@kch.hosp.go.jp)
Yasushi Morioka (shayashi@kch.hosp.go.jp)
Kazuyoshi Hatsuda (shayashi@kch.hosp.go.jp)
Eriko Naito (shayashi@kch.hosp.go.jp)
Mitsunori Sakatani (shayashi@kch.hosp.go.jp)

Version: 3 Date: 18 December 2009

Author's response to reviews: see over
Covering letter
Dear Sir.,

We appreciate your review of this work, and I apologize for the delay of submission. We are enclosing a manuscript entitled “Prompt improvement of pressure ulcer and tuberculosis by semisolid nutrition with high viscosity given by nasogastric tube; a case report” for publication in Journal of Medical Case Reports.

According to your indications
1. Race and nationality of the patient were specified in the text (page 2, line 9).
2. Figures were deleted from the manuscript.
3. Figure cropping was conducted according to figure guide.
4. The two figures as separated figure files were uploaded using the “upload” system.

Sincerely yours,

Seiji Hayashi, MD.,PhD.
Kinki-chuo Chest Medical Center